Author's response to reviews

Title: Gastric Emptying is Slow in Chronic Fatigue Syndrome

Authors:

Richard B Burnet (rburnet@mail.rah.sa.gov.au)
Barry E Chatterton (barry.chatterton@adelaide.edu.au)

Version: 5  Date: 16 December 2004

Author's response to reviews:

The authors have included all suggested revisions in this final version.
Capitalisation changed
hyphen removed
Authors qualifications removed
Authors names in order
Affiliation and address format met
Conclusions included
keywords all in text and abstract and deleted from manuscript.
Capitalisation removed.
Punctuation requests complied with
Competing interest and author's contribution section added.
Tables in requested format.
Titles removed from figures, and white space cropped